Author’s response to reviews

Title: Prevalence of Intestinal Parasitic Infection and Associated Risk Factors among Schoolchildren in Gondar University Community School, Northwest Ethiopia: a cross sectional study

Authors:

Aschalew Gelaw (aschalew3@gmail.com)
Belay Anagaw (belayanagaw@yahoo.com)
Bethel Nigussie (bethel.nigussie@yahoo.com)
Betrearon Silesh (betrearon8@gmail.com)
Atnad Yirga (attnadn@gmail.com)
Meseret Alem (mesey4839@yahoo.com)
mengistu Endris (mengist06@yahoo.com)
Baye Gelaw (bgt2@leicester.ac.uk)

Version: 2 Date: 13 September 2012

Author’s response to reviews: see over
Cover letter

To: BMC public health editorial office

Subject: our manuscript

We the authors of the manuscript entitled “Prevalence of Intestinal Parasitic Infection and Associated Risk Factors among Schoolchildren in University of Gondar Community School, Northwest Ethiopia: a crosssectonal study” decided to publish our manuscript on your reputable journal. We want to submit our work to BMC public health is because of its accessibility, reachable to reader, fastest response. We declare that this work is fully our work and yet not published or submitted for publication to any other journals.

With regards

Aschalew gelaw

Corresponding author